Author's response to reviews

Title: SMS and phone call: method used to motivate the presence appointment of children infected or exposed to HIV, study protocol for a multisite factorial randomized controlled trial (SMS-Call Children)

Authors:

Jean Joel Bigna Rim (bignarimjj@yahoo.fr)
Charles Kouanfack (charleskouanfack@yahoo.fr)
Sinata Koulla Shiro (koullasinata@yahoo.fr)

Version: 3 Date: 24 June 2013

Author's response to reviews:

Cover Letter: SMS-Call Children

Good morning.

The reasons that will incite you to publish this article are:

1. The protocol presents a pioneering study on the use of SMS and phone calls for medical appointments in Africa.
2. The protocol has a randomized factorial trial, which is not the case for previous studies.
3. The study population is in the field of pediatric HIV, not previously considered in the context of telemedicine.

Best regards.